\begin{table}%t2
\caption{The coefficients $a_i$ and $b_i$ in Eq.~(\ref{eq:Fernie:other}).}
\label{tab:Fernie:other}
%\centering
\par
\begin{tabular}{llrrcr}
\hline
\hline
 \multicolumn{1}{c}{Authors} & &
 \multicolumn{1}{c}{$a_i$} & \multicolumn{1}{c}{$b_i$} &
 \multicolumn{1}{c}{$\sigma$} & \multicolumn{1}{c}{$n$} \\
\hline
Laney \& Stobie (\cite{Laney:Stobie:93})  & LS   & $-0.090$ & $ 0.019$ & $0.05$ & $ 48$  \\
Dean et~al. (\cite{Dean:etal:78})     & DWC  & $-0.012$ & $ 0.017$ & $0.05$ & $ 95$  \\
Parsons \& Bell (\cite{Parsons:Bell:75})  & PB   & $-0.047$ & $-0.012$ & $0.07$ & $158$  \\
Turner et~al. (\cite{Turner:etal:87})   & TLE  & $ 0.022$ & $-0.002$ & $0.04$ & $ 26$  \\
Dean (\cite{Dean:81})          & DE   & $ 0.041$ & $-0.001$ & $0.02$ & $ 36$  \\
Pel (\cite{Pel:78})           & PE   & $ 0.003$ & $ 0.010$ & $0.04$ & $ 84$  \\
Eggen (\cite{Eggen:96})         & EG2  & $ 0.177$ & $-0.043$ & $0.05$ & $ 55$  \\
Yakimova et~al. (\cite{Yakimova:etal:75}) & YT   & $ 0.047$ & $-0.022$ & $0.05$ & $152$  \\
Bersier (\cite{Bersier:96})       & BERS & $-0.033$ & $-0.003$ & $0.04$ & $ 24$  \\
\hline
\end{tabular}
\end{table}